**Author's response to reviews**

**Title:** Behavioral intentions in response to an influenza pandemic

**Authors:**

Gerjo Kok (g.kok@maastrichtuniversity.nl)
Ruud Jonkers (r.jonkers@rescon.nl)
Roger Gelissen (r.gelissen@rescon.nl)
Ree Meertens (r.meertens@maastrichtuniversity.nl)
Herman P. Schaalma (g.kok@maastrichtuniversity.nl)
Onno de Zwart (dezwarto@ggd.rotterdam.nl)

**Version:** 5  **Date:** 16 March 2010

**Author's response to reviews:** see over
Dear Editors,

We have further improved the manuscript in response to the comments made by the reviewer. We also have gone through the manuscript formatting checklist one more time and have tried to ensure that our revised manuscript conforms to all of the points.

Reviewer: Ivar Sonbo Kristiansen:
1. Remove items from the questionnaire if they are not used for the paper
2. Clarify the text for Figure 1. Do the circles in the graph represent mean responses in the current study - or what?

1. We deleted questions 47-50 from the questionnaire as they were not part of the paper.
2. We added “(the dots represent mean responses)” to the title of figure 1

Best regards, Gerjo Kok